Author's response to reviews

Title: Identifying patient preferences for communicating risk estimates: a descriptive pilot study

Authors:

Jennifer M. Fortin (fortin@dsgr.harvard.edu)
Linda K. Hirota (lhirota@alumni.tufts.edu)
Barbara E. Bond (bbond@lifespan.org)
Annette M. O'Connor (aoconnor@lei.ca)
Nananda F. Col (ncol@dsgr.harvard.edu)

Version: 5  Date: 25 Jul 2001

Dear Ms. King,
Per your request, we have resubmitted our manuscript with the following revisions:
1. Included "Competing Interests" section
2. Revised title of manuscript and capitalized in English style
3. Added an author affiliation for Linda Hirota
4. Added a country to each author's affiliation
5. Reformatted reference list to be consistent with Biomed Central style
6. Improved visibility of Figures 1 and 2 by breaking each figure up into 2 sections (1a, 1b and 2a, 2b respectively).

Thank you for your helpful suggestions. If you need any additional information, please do not hesitate to contact Dr. Nananda Col at 617-732-9203, ncol@dsgr.harvard.edu. We look forward to hearing back from you.